



# Article Predicting Activity Duration in Smart Sensing Environments Using Synthetic Data and Partial Least Squares Regression: The Case of Dementia Patients

Miguel Ortiz-Barrios <sup>1,\*,†</sup>, Eric Järpe <sup>2,†</sup>, Matías García-Constantino <sup>3,†</sup>, Ian Cleland <sup>3,†</sup>, Chris Nugent <sup>3,†</sup>, Sebastián Arias-Fonseca <sup>1,†</sup> and Natalia Jaramillo-Rueda <sup>1,†</sup>

- <sup>1</sup> Department of Productivity and Innovation, Universidad de la Costa CUC, Barranquilla 08002, Colombia; sarias9@cuc.edu.co (S.A.-F.); njaramil@cuc.edu.co (N.J.-R.)
- <sup>2</sup> Department of Intelligent Systems and Digital Design, Halmstad University, P.O. Box 823, S 301 18 Halmstad, Sweden; eric.jarpe@hh.se
- <sup>3</sup> School of Computing, Computer Science Research Institute, Ulster University, Belfast BT37 0QB, UK; m.garcia-constantino@ulster.ac.uk (M.G.-C.); i.cleland@ulster.ac.uk (I.C.); cd.nugent@ulster.ac.uk (C.N.)
- \* Correspondence: mortiz1@cuc.edu.co
- + These authors contributed equally to this work.

Abstract: The accurate recognition of activities is fundamental for following up on the health progress of people with dementia (PwD), thereby supporting subsequent diagnosis and treatments. When monitoring the activities of daily living (ADLs), it is feasible to detect behaviour patterns, parse out the disease evolution, and consequently provide effective and timely assistance. However, this task is affected by uncertainties derived from the differences in smart home configurations and the way in which each person undertakes the ADLs. One adjacent pathway is to train a supervised classification algorithm using large-sized datasets; nonetheless, obtaining real-world data is costly and characterized by a challenging recruiting research process. The resulting activity data is then small and may not capture each person's intrinsic properties. Simulation approaches have risen as an alternative efficient choice, but synthetic data can be significantly dissimilar compared to real data. Hence, this paper proposes the application of Partial Least Squares Regression (PLSR) to approximate the real activity duration of various ADLs based on synthetic observations. First, the real activity duration of each ADL is initially contrasted with the one derived from an intelligent environment simulator. Following this, different PLSR models were evaluated for estimating real activity duration based on synthetic variables. A case study including eight ADLs was considered to validate the proposed approach. The results revealed that simulated and real observations are significantly different in some ADLs (*p*-value < 0.05), nevertheless synthetic variables can be further modified to predict the real activity duration with high accuracy ( $R^2$  (*pred*) > 90%).

**Keywords:** activities of daily living (ADLs); activity recognition; activity duration; partial least square regression (PLSR); people with dementia (PwD); simulated data; artificial intelligence; sensor systems; smart homes

## 1. Introduction

Recent advances in medicine and healthcare have contributed to the increase in life expectancy worldwide, which has also brought more incidence of neurodegenerative conditions, such as dementia, in the elderly population. As reported by the World Health Organisation (WHO) in 2022, there are more than 55 million people living with dementia around the world, with an increase of nearly 10 million cases per year [1]. While the COVID-19 pandemic brought wider public attention to the challenges that health systems and carers (formal and informal) of people with dementia (PwD) face, these challenges were already acknowledged before the pandemic [2]. Technological means in the form of



**Citation:** Ortiz-Barrios, M.; Järpe, E.; García-Constantino, M.; Cleland, I.; Nugent, C.; Arias-Fonseca, S.; Jaramillo-Rueda, N. Predicting Activity Duration in Smart Sensing Environments Using Synthetic Data and Partial Least Squares Regression: The Case of Dementia Patients. *Sensors* **2022**, *22*, 5410. https:// doi.org/10.3390/s22145410

Academic Editors: Marco Mobilio and Daniela Micucci

Received: 15 June 2022 Accepted: 13 July 2022 Published: 20 July 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). mobile apps and sensors to monitor the activities of daily living (ADLs) of PwD detect their behaviour patterns and emergency situations that require effective and timely human interventions have been more widely used to support their carers with good results [3].

Sensor-based ADLs monitoring solutions typically perform well in controlled laboratory conditions; however, in many cases, these solutions do not perform as expected in real-world conditions, such as in care homes [4], mainly because of uncertainties derived from smart home configurations and the different ways in which people perform ADLs. In this sense, while collecting real-world data might allow the development of more personalised approaches for each individual, this data collection is usually costly and characterized by a challenging recruiting research process, in which PwD may not be willing to adopt technological solutions [5]. In many sensor data collection instances that are part of research projects, there is usually not sufficient data to be analysed in such a way that has personalised benefits for the PwD involved. The use of synthetic data generated through simulation has been more frequently used as an alternative efficient choice to complement real datasets [6], overcoming the lack of sufficient real-world data and providing the capability of producing as much synthetic data as required for machine learning algorithms to generate personalised solutions. However, synthetic data can be significantly dissimilar compared to real data in the sense that while the generated data may be proportional and within acceptable ranges in general, they might not be an accurate and realistic representation of the specific cases of the PwD aimed at. Hence, the synthetic data would be an inaccurate alternative to data otherwise collected for each PwD and could lead to misleading and erroneous conclusions about the types, frequency, and duration of ADLs performed by PwD.

This paper proposes the application of Partial Least Squares Regression (PLSR) to approximate the real activity duration of various ADLs based on synthetic observations. The approach presented involves two phases: (i) comparing the duration of the ADLs that were collected from PwD (real data) with data derived from an intelligent environment simulator (synthetic data), and (ii) evaluating different PSLR models to estimate real activity duration of synthetic data. The criteria considered to estimate the real activity duration are: (i) number of events per activity (NEPA), (ii) number of events per sensor per activity (NEPSA), and (iii) activity duration. A case study is presented to evaluate this and it considers eight ADLs, which are deemed to be common activities performed by most people: (i) stay in bed, (ii) use restroom, (iii) make breakfast, (iv) get out of home, (v) get cold drink, (vi) stay in the office, (vii) get hot drink, and (viii) cook dinner. Note that while for healthy people without disabilities performing the ADLs considered may be easier to perform than for PwD, a machine learning personalised approach would be of benefit for both types of people in the prediction of activity duration, which in many cases can be related to health problems that could worsen if unattended, such as neurodegenerative diseases or infections. For example, inaccurate synthetic data about the frequency and duration of the "use of bathroom" ADL could present a higher frequency and a large duration that could then result in the carer or nurse of the PwD increasing the use of a drug for Urinary Tract Infection (UTI).

The main research question of our investigation is if statistical methods, such as PLSR, can be used on synthetic data to reliably approximate and predict the duration of ADLs performed by PwD. Therefore, the novelty of our approach is that it can use synthetically generated data to provide accurate personalised predictions on ADLs performed by PwD, significantly reducing the time in which a prediction model would be generated using just real-world data. This is because in the case of real-world data, it typically takes longer and involves more costs (staff salaries, participants' recruitment, and use of equipment) to perform the collection, pre-processing, and formatting of the data so it can be used by statistical methods to then obtain predictions.

The research contributions of this paper are: (i) producing predictive models that provide a more accurate transformation of simulated data to describe duration of real activities, and (ii) identifying the main synthetic predictors for real activity duration in each ADL and its intrinsic properties, and how this metric varies amongst users. This study aims to bridge gaps of previous research works in contrasting real and synthetic data, for instance, the SynSys algorithm proposed by [7] does not take into account the nature of each ADL, which our approach does. It is intended that the model will be improved over time as more real-world data is collected for each PwD while providing a solution based on synthetic data that can provide early benefits to PwD. In practice, our approach would also reduce the costs involved in using ambient and wearable sensors for extended periods of time to collect real-world data from PwD, as well as costs involved with sensor maintenance, and data collection, pre-processing, and analysis.

The remainder of this paper is organised as follows: Section 2 presents the related works on ADLs and the generation of synthetic data. Section 3 presents the materials and methods used. The experiment definition is described in Section 4. Section 5 presents the results obtained and the discussion. Finally, Section 6 presents the conclusions.

#### 2. Related Works

The ability to accurately recognise activities is central to many intelligent systems including smart home automation, ambient assisted living, assistive robotics and human computer interaction. For people living with dementia, activity recognition (AR) techniques have facilitated several use cases including detection of agitation [8], tracking cognitive impairment [9], detecting anomalous activities [10], detecting urinary tract infections [11], and mapping social interactions. This section provides a brief overview of the challenges associated with activity recognition within smart environments before discussing the potential of synthesised data to address this.

#### 2.1. Synthesizing Data for Sensor Based Activity Recognition

AR research can be categorised into two main approaches; namely video or sensorbased. Sensor-based AR lags similar fields largely due to a lack of large-scale, highquality, multi-modal, and labelled datasets. This has impeded progress in developing robust and generalised Machine Learning (ML) approaches. The success of supervised ML relies primarily on the availability of large datasets with high-quality annotations. Collecting labelled data or employing experts to label large datasets are infeasible in resource-constrained settings, such as healthcare. Indeed, the collection of human activity and behaviour data is time consuming, costly, and often limited in terms of availability.

To further complicate this issue, owing to biological and environmental factors, the same activity can be performed differently by different individuals. These differences are further enhanced when considering the variation that occurs in those with movement disorders (cerebral palsy, post-stroke) or cognitive impairment (dementia). Despite the amount of research undertaken in the field, issues such as cross-subject variability are still posing an obstacle to the deployment of solutions in large-scale and free-living settings.

To address these problems, researchers have been investigating the use of synthetic data for training, testing, and validation of ML-based AR techniques [7]. Early research in this area typically relied upon mathematical models such as Markov chains or Petri networks [12]. This then progressed to investigate combinations of approaches to model more complex activities. For example, Helal et al. [13] developed a solution using Markov chains to model activity patterns and combined this technique with Poisson distribution to generate realistic timestamps. This work later progressed to a software solution to allow researchers to create synthetic datasets.

More recently, researchers have been investigating the use of Generative Adversarial Networks (GANs) to generate synthetic sensor data. This approach utilises an adversarial discriminative model to determine whether a generated synthetic sample follows the same distribution as real data. The generative model continuously improves the quality of the synthetic data until the adversary is unable to distinguish real from synthetic data. Whenever large amounts of labelled data are unavailable, synthetic data can be generated to augment the available labelled data to provide enough data for training and testing.

Using GANs, Alharbi et al. [14] explored and constructed a model for generating several types of human activity sensor data. They assessed the use of synthetic data to train two commonly-used classifier models, Convolutional Neural Network (CNN) and Long Short Term Memory (LSTM). In doing so, they demonstrated the efficacy of the proposed method on two publicly available human activity datasets, the Sussex-Huawei Locomotion (SHL) and Smoking Activity Dataset (SAD). The solution achieved improvements for both SHL (0.85 to 0.95 F1-score), and for SAD (0.70 to 0.77 F1-score) when using a CNN activity classifier.

## 2.2. Challenges with Existing Approaches for Data Synthesis

As summarised above, existing approaches for synthesising realistic data have shown great promise in improving the accuracy of ML approaches for AR. An ongoing challenge, however, in developing such solutions is the ability to generate simulated data that accurately represents real data. A comparison between real data collected within the Gator Tech Smart House and simulated data generated by Persim 3D [15] revealed average data similarities of between 78% and 81%. Another study comparing real data with data generated using the simulator MASSHA found the similarity to be between 88.10% and 93.52% in terms of frequency, and 98.27% and 99.09% in terms of duration on datasets containing single user activities [16].

Smart home data has a number of unique characteristics in comparison to most sequence and time-series data. Sensor data for activity recognition are typically not independently and identically distributed and exhibit both high dimensionality and a high level of complexity. Therefore, synthetic data must be generated that are consistent with these characteristics. Firstly, within a smart home, data do not typically arrive at a constant rate. Therefore, synthetic data must include realistic time stamps for each sensor reading. Secondly, many smart home sensors do not produce continuous values. Therefore, when synthesising data, a sensor name must be generated along with a corresponding value. For light or temperature sensors, the corresponding value is numeric at a set sample rate. For motion or door sensors, the corresponding activity labels. In real experiments, activity labels are provided by external annotators. The label must therefore also be generated and attributed to the correct set of sensors, activating in the correct sequence and over the right time frame.

To address these complexities, Dahmen and Cook [7] developed SynSys. This solution semi-supervised learning combined with synthetic data generation to generate synthetic data could be used to improve the accuracy of ML approaches for AR. They showed that the SynSys algorithm was able to successfully generate more realistic synthetic data for a week of smart home data in comparison to random data, data from another home, and synthesised data created using a single Hidden Markov Model (HMM) and Poisson regression technique. Combining real data with synthetic data generated from SynSys was shown to significantly improve the accuracy of activity recognition. The authors also noted that as SynSys is based on generating nested sequences of data and sequences of timestamps that capture duration, the method can be applied to more generalized data beyond the smart home datasets we used in our experiments.

Based on the existing literature, it has been noted that research comparing real observations and simulated data is largely limited and poorly developed. Alshammari et al. [17] reviewed 228 software tools for simulation and found that only a small number focused on generating datasets with the majority being focused on visualisation and context-awareness applications. Similarly, Bouchabou et al. [18] highlighted that whilst simulation tools provide an excellent opportunity to quickly generate and visualise smart home date; in particular, it is noted that synthetic datasets allow for quick evaluation of different sensor configurations within the environment without the requirement of physical deployment and volunteer subjects. Additionally, the annotation can be more precise compared to real dataset methods. Nonetheless, synthetic data can be un-realistic in comparison to

real-world datasets. Typically, activities provided within synthetic datasets are less realistic in terms of execution rhythm and variability. Furthermore, the design of the virtual smart homes can be cumbersome for a non-expert designer. Bouchabou et al. [18] also noted that there are currently no publicly available synthetic datasets, though a number of open simulators, such as OpenSH, are available. In spite of the strong demand from ML surprisingly little has been done about assessing the validity of simulation-based studies and methodology for real data problems in image analysis and computer vision [19] and in a more general perspective [20]. In fact, Kleijnen [21] argues that measurements for verification and validation of simulation models are in need of development.

As reported in this section, several approaches (Markov chains, software solutions like the PerSim3D and the MASSHA activity simulators, GANs, and SynSys) have addressed the generation of synthetic data with different levels of success in relation to how close the synthetic data was with real data as measured using different metrics. While the authors of the related works presented in this section highlight positive results, it is necessary to have a more standardised way (benchmark datasets, use same metrics, etc.) to determine their effectiveness and their limitations. This paper bridges this gap by (i) evaluating the equivalence between real and synthetic activity duration, and (ii) implementing PLSR to better approximate real activity duration based on simulated observations.

#### 3. Materials and Methods

The data used in this study consists of actual observations of duration times for different daily activities and synthetic data mimicking the probabilistic properties of the real-world data. Please, refer to Section 4 for more information about the experiment setup and data used for the study.

The traditional statistical tool for establishing dependence structure between random variables in terms of their explicit (linear or non-linear) relationships is regression analysis, commonly carried out having estimated the model parameters by means of ordinary least squares regression (OLSR). However, in cases when the sample sizes are small, there are missing data and/or in the presence of strongly correlated variables, a development of the regression procedure is to base the analysis on Partial Least Squares (PLS). This initiative was taken by Wold [22] more than half a century ago and has proven an interesting idea [23–28] to come to terms with the numerous possible problems with regression based on OLS. There has been much development, [29–34] to mention a few, but also critical papers, e.g., [35–37]. There are also other approaches to cases with, e.g., imbalanced data. For example, for the detection of voice disorders in an imbalanced dataset, General Adversarial Network (GAN) and C-means clustering were used [38]. Nevertheless, the many problems with GANs are a well documented fact (see, e.g., [39,40]) mainly related to poor convergence of model parameters. Additionally, the cumbersome loss function implies that the model is very hard to train [41]. Thus, GAN is considered to be less appropriate for simulating activity data among people with dementia. Among Recurrent Neural Networks (RNNs) [42] are models and techniques for including time dependent data. By utilizing memory and so-called gates the Long-Short Term Memory Neural Network (LSTMNN) and the Gated Recurrent Unit (GRU) are methods which can learn how to filter out irrelevant data and remember the relevant. A problem here though is to optimally choose initial values for the weights of the oblivion gate and for the memory gate. Another disadvantage with these models is that the training needs very large amounts of data. PLS and its variants are to be considered as fast solutions [43] for parameter estimation, here, for simulation objectives. However, there are other methods of Machine Learning (ML) which can adapt to the data and produce simulations which more accurately correspond to the real data from a probabilistic point of view. Still, the price to pay is a higher computational complexity [44], more time-consuming simulation procedure, and sometimes convergence issues. For example, automatic pruning methods have been employed to reduce the complexity of AI models with multiple layers and a high number of parameters which are very difficult to deploy on resource-restricted platforms [45]. Similarly, the quantization approach has been proposed

for shortening the number of trainable parameters and the associated computational effort of the entire implementation [46]. As previously discussed, the aforementioned methods are very useful for addressing overparametrized data structures. Nonetheless, in this case study, we only consider three features and their combinations which may not represent a high computational complexity. Indeed, PLSR is recognized by virtue of its computational efficiency [47] with applications in remote sensing environments [48] as exposed in this research. In ML, the complexity of methods is rarely below a polynomial time while in PLSR the complexity of methods is typically below the polynomial time. With PLSR the gain is a robust method with quick delivery. Below follows a very brief introduction of PLS, but a more thorough presentation of the field can be found in, e.g., [26–28].

The notion of partial least squares regression (PLSR) includes multiple regression, principal components analysis (PCA), and principal components regression (PCR) [28]. PCA allows pinpointing the smallest number of uncorrelated features, which reduces the complexity of the prediction model. Linear PLSR without intercept may be formally presented as follows. In Equation (1), the model is specified. For given *n* observations of covariates, *X*, (*m* variables) and *n* corresponding observations of responses, *Y*, (which can be *k* variables) the object is to minimize  $||\epsilon||$  such that

$$Y = TBQ^T + \epsilon \tag{1}$$

where  $X = TP^T + \epsilon_1^T$ ,  $Y = UQ^T + \epsilon_2^T$ , X is an  $n \times m$  matrix, Y is an  $n \times k$  matrix, T is an  $n \times a$  matrix, P is an  $m \times a$  matrix, U is an  $n \times a$  matrix, Q is an  $m \times a$  matrix, B (regression coefficients),  $\epsilon$ ,  $\epsilon_1$ ,  $\epsilon_2$  (residuals) are  $n \times 1$  vectors,  $\|\cdot\|$  is the Euclidean norm and  $\cdot^T$  denotes the matrix transpose. Determination of T, P, U and Q may be achieved by deploying the Nonlinear Iterative Partial Least Squares (NIPALS) method. This is an iterative procedure that converges to the optimal solutions for the least squares problem, which minimizes  $\|\epsilon\|$  under the condition either of leaving out column vectors of T and U which do not correspond to variables with small eigenvalues or by systematically, for each iteration in the procedure, carry out an Analysis of Variance (ANOVA) based on resampling by means of a bootstrap method, which determines whether all coefficients in the model are significantly non-zero or not. Regarding the generalization ability, Hao and Chen [49] and Zheng et al. [50] proposed an interesting use of transferable feature learning and instancelevel adaptation for visual recognition; in this case, the transferability of general features is achieved by calculating the regression coefficients *B* of each ADL while replacing the covariates X by the corresponding synthetic observations derived from the simulator when interacting with the user.

#### Assessment of Model and Data

For ordinary least squares regression assumptions such as normality of residuals and independence between covariates should be checked. Regarding corresponding assumptions for PLSR one should check independence and equal distribution of the residuals. Independence is here examined with the Durbin–Watson test by rejecting for large values of the statistic in Expression (2)

$$\frac{\sum_{i=2}^{n} (\epsilon_i - \epsilon_{i-1})^2}{\sum_{i=1}^{n} \epsilon_i^2} \tag{2}$$

according to the  $\chi^2$ -distribution with n - 1 degrees of freedom. The determination coefficient  $R^2$  is calculated for learning about how much of the variation in the response that is explained by the model. Additionally, the Prediction Residuals Sum of Squares (PRESS) [51] as presented in Expression (3)

$$\operatorname{diag}\left((Y-\hat{Y})^{T}(Y-\hat{Y})\right)$$
(3)

is calculated to give some idea about the predictive quality of the model where  $\hat{Y}$  is the observed response and  $\hat{Y}$  is the predicted response according to the model. This evaluation

measure is a common indicator of the optimal number of covariates. However, it is sometimes considered to be awkward due to the very large number of calculations required to be determined [52,53].

Apart from modelling, there is also a need for methodology to distinguish potential differences between real and synthetic data. To this end, a paired sample Wilcoxon test is performed for each pair of real and synthetic activity variables [54]. For the variables  $X_1$  and  $X_2$ , the hypothesis as in Expression (4)

$$\begin{cases} H_0: p = \frac{1}{2} \\ H_1: p \neq \frac{1}{2} \end{cases}$$
(4)

is to be tested where  $p = P(X_{1,i} + X_{1,j} > X_{2,i} + X_{2,j})$ ,  $X_{1,1}, X_{1,2}, \dots, X_{1,n}$  are observations of  $X_1$  and  $X_{2,1}, X_{2,2}, \dots, X_{2,n}$  are observations of  $X_2$ . Thus, the statistic

$$\Gamma^+ = \#\{X_{1,i} + X_{1,j} > X_{2,i} + X_{2,j} : i, j = 1, \dots, n \text{ and } i \le j\}$$
(5)

is calculated as given in Equation (5). This is rejected for values  $T^+ = t$  such that  $F_{T^+}(t) > 1 - \frac{\alpha}{2}$  or  $F_{T^+}(t) < \frac{\alpha}{2}$  for a size  $\alpha$  test. Here,  $F_{T^+}$  denotes the approximate cumulative distribution function (cdf) [55]

$$F_{T^{+}}(t) \approx \Phi(\omega) + \frac{(3n^{2} + 3n - 1)\omega(\omega^{2} - 3)}{10n(n+1)(2n+1)} \cdot \frac{d\Phi(s)}{ds}\Big|_{s=\omega}$$
(6)

of  $T^+$  as given in Equation (6) where Equation (7) specifies the notation

$$\omega = \frac{4t + 2 - n(n+1)}{\sqrt{\frac{8}{3}n(n+1)(2n+1)}}$$
(7)

and  $\Phi(s)$  is the standard normal cdf. For further information about all these methods, please consider standard texts such as [56,57].

## 4. Experiment Definition

The real environment is the Halmstad Intelligent Home (HINT) located in Halmstad, Sweden [58,59], which is supplied with contact/touch sensors recognizing occupancy in chairs and sofas. This apartment is also provided with Passive Infrared (PIR) sensors located in different spaces, thereby facilitating the continuous monitoring of motion and occupancy of subjects staying in their settings. Likewise, there are switches detecting the opening/closing of different home appliances and furniture doors installed in this home. The real architecture and floor plant layout of HINT are depicted in Figure 1. HINT is then suitable for underpinning two main purposes: (i) parsing out the ADLs performed by PwD over time and (ii) designing anticipated interventions for assisting PwD in case of severe health decline.



Figure 1. (a) The HINT layout. (b) The sensing capabilities of HINT.

HINT was virtually modelled by IE Sim, a flexible interactive simulation software developed by Synnott et al. [60] considering the current sensor network placement within the home and the floor plan (Figure 1). The simulator also emulated the furniture items available in the real environment to make the representation more realistic and then collect activity duration data with better approximation to the real-world pattern.

Whilst a huge amount of research has been carried out in the area of sensor-based human activity recognition, there is a lack of standardization in terms of methodology, particularly around the types of activities investigated and how performance of models is evaluated [61]. Currently, there is no recognised taxonomy of activities that should be investigated [18]. However, many previous works have focused on ADLs as described by Katz [62] to define a list of basic and necessary activities. Considering the lack of a standard taxonomy for activities including bathing, dressing, toileting, transferring, and feeding. As part of the experiment, eleven subjects were requested to undertake 8 ADLs (Stay in bed, Use restroom, Make breakfast, Get out of home, Get cold drink, Stay in the office, Get hot drink, and Cook dinner) while staying at HINT (Figure 2). Additionally, the same participants were asked to perform the ADLs using the IE Sim simulator. The following list of instructions was provided to guide the user on the sequence of tasks during the simulation:

## Previous general commands

- Please carefully read the tasks to be carried out in each activity.
- Please feel free to re-read the instructions of each activity if necessary.
- Please press the Start/Stop switch when an activity is finished.
- Please lock the door upon crossing through.
- Please switch off each household appliance after use.

#### ADLs description

## ADL 1: Stay in bed

You can remain in bed as long as you wish. The maximum time is 2 min. After this, you have to get out of the bedroom, lock the door, and press the Start/Stop switch.

# ADL 2: Use restroom

You can use the hand-washing sink and/or toilet if you need. After this, please get out of the bathroom, lock the door, and press the Start/Stop switch.

#### ADL 3: Make breakfast

You have to cook something for breakfast. Besides, you can select between milk and cereals or coffee. However, it is also possible to prepare both if you want. After this, move the bowl up on the dining table, sit down, and press the Start/Stop switch.

## ADL 4: Get out of home

You can decide to leave the house from the courtyard door or from the front door. When you are in outdoors, please push the Start/Stop switch.

#### ADL 5: Get cold drink

You can take the drink from the refrigerator or serve plain water. After this, put the poured glass on the kitchen table and press the Start/Stop button.

## ADL 6: Stay in the office

Please proceed to the office and push the Start/Stop button.

#### ADL 7: Get hot drink

You can select between preparing coffee or tea. After this, put the poured cup on the kitchen table and push the Start/Stop button.

# ADL 8: Cook dinner

Please make soup. Put the served bowl on the kitchen desk and push the Start/Stop button.



(a) PHOTO: Halmstad University

(b) PHOTO: Anders Andersson

**Figure 2.** (a) A participant undertaking some of the ADLs in the kitchen and bedroom. (b) Human behaviour monitoring at HINT.

The results were two datasets (synthetic and real) containing a compilation of sensor events per ADL with each related to a time mark, the activated sensors (e.g., contact/touch, PIR, opening/closing), sensor identification code, and status (open or closed). Following this, three indicators were estimated: ADL duration, number of events per ADL, and number of events per sensor type per ADL. Previously, the datasets were cleaned by removing outliers deriving from errors during the execution of the experiments, thereby increasing the fit and predictive ability of PLSR models depicted in next section.

## 5. Results and Discussion

The ADLs performed by the subjects while staying at HINT are represented by more than 900 real sensor activations ( $\mu = 1.93$ ;  $\sigma^2 = 3.45$  min per subject). On the other hand, approximately 1100 synthetic sensor events were derived from interactions with the virtual IE Sim environment ( $\mu = 1.675$ ;  $\sigma^2 = 13.80$  min per participant). Overall, 36 synthetic samples (10 activity durations, 10 events per activity, and 16 events per sensor per activity) were generated per each ADL. This represents a total of 288 samples in the whole experiment. Prior to undertaking the ADLs in IE Sim, the subjects first learned how to handle the software to diminish bias and flatten the learning curve. In this case, all the participants completed the ADLs satisfactorily in both real and synthetic homes. In fact, the average time spent by each subject carrying out all the ADLs was 8.69 ( $\sigma^2 = 4.21$  min) and 13.93 ( $\sigma^2 = 12.65$  min) at HINT and IE Sim, respectively. When comparing these metrics, it is evident that, in general, the time used by participants to complete all the ADLs in the simulator is significantly higher than the one spent in the real apartment (T-value = 4.24; p-value = 0.002; 95% CI for the difference: [2.49–7.99] min). The following sub-section will elucidate if this difference is also detected in all the ADLs individually. PLSR models will be applied to better approximate the simulated data to real activity durations in those ADLs with significant differences.

## 5.1. Contrasting Real and Synthetic Activity Duration

A two-sample Wilcoxon test ( $\alpha = 0.05$ ) was applied to analyse the equivalence between real and synthetic activity duration in each of the eight ADLs defined in the experiment description. Thereby, it will be possible to detect those ADLs in which a statistical transformation is needed to better approximate the synthetic activity duration to the one in the real environment. Table 1 enlists the results of the comparison in terms of *p*-values, 95% confidence interval (CI) for the median difference, and the conclusions. The results are also graphed through individual value plots comparing the data distribution and medians between the synthetic and real observations (Figure 3).

10 of 21

ADL	<i>p</i> -Value 9	5% CI for the Difference (sec)	Conclusion
Stay in bed	0.093	[-5: 100]	Statistically similar
Use restroom	0.050	[-34; -1]	Statistically different
Make breakfast	0.012	[-66; -22]	Statistically different
Get out of home	0.889	[-12.16]	Statistically similar
Get cold drink	0.161	[-36:7]	Statistically similar
Stav in the office	0.101	[-104:64]	Statistically different
Cat hat drink	0.012	[ 236, 62]	Statistically different
Get hot utilik	0.010	[-230, -02]	Statistically different
Cook diffiler	0.012	[-139, -44]	Statistically different
200		100 -	•
	•	90.	
	٠	0 80	
• ti		• <sup>10,</sup> tio	•
E 100	107.5	00 III	•
	<u> </u>	रू 50 <b>.</b>	49.5
vit	•	40. 39 S	
<sup>50</sup> 43	•	itj 30.	•
₹	•	20.	
•		•	
Sim Stay in Bed	Real Stay in Bed	Sim Use restro	oom Real Use restroom
_ ,	_ ,	_	-
130.	•	70	
120			•
Og 110.	:	9 <sup>60</sup>	
⊆ 100	101.5		
e ti	-	·£ 50	
• Internet	•	nra	•
р х q	•		
I I I I I I I I I I I I I I I I I I I	•		•
		• 30	\$32.5
50. 54			••
40 8		20	•
Sim_Make breakfast	Real_Make breakf	ast Sim_Get out of h	nome Real_Get out of_home
140		140	
•		140	•
G <sup>120</sup>		G <sup>120</sup>	•
ěs) 100		Š 100	<b>9</b> 99 5
u o			
<sup>08</sup> atic	•	<sup>08</sup> atio	•
•	• • • • • •		
•	60.5	± 100 m	
40. 43.5	•	40	
¥		A A A	
20		15.5	
ol			
Sim_Get cold drink	Real_Get cold dri	nk Sim_Stay in the	office Real_Stay in the office
350	•	250	
200	•		•
$\overline{\mathfrak{O}}^{300}$	•	2 200	•
<u>s</u> 250.	•	(se	
u o o		.0 150	<u>•</u>
	185		137.5
P 150.			•
	•	• [jui ]	•
	•		· •
₹ 50. 55		₹ <sup>50</sup> 40	÷
		•	
Sim Get hot drink	Real Get hot drin	 Ik Sim Cook din	ner Real Cook dinner

 Table 1. Two-sample Wilcoxon test results.

**Figure 3.** Differences between synthetic and real activity duration. The ADLs in the first row from the left are: *Stay in bed* and *Use restroom;* second row: *Make breakfast* and *Get out of home;* third row: *Get cold drink* and *Stay in the office;* and fourth row: *Get hot drink* and *Cook dinner.* 

The Wilcoxon test outcomes revealed significant differences (p value  $\leq 0.05$ ) between the simulated and real-world data for the activity duration indicator in five ADLs: *Use restroom, Make breakfast, Stay in the office, Get hot drink,* and, *Cook dinner*. It is then concluded that the simulator does not accurately represent the real intrinsic properties of users when performing these activities. This is also confirmed by the 95% confidence intervals (CI), which do not include zero, but show lean to the left. Mostly, the gaps have been detected in activities with a larger number of sensor events than those performed in the kitchen.

Looking into the results, it is evident that simulated activity duration data tend to be meaningfully minor compared to those from the real smart home. In this regard, it is important to identify the sensors triggering the gap as well as analysing the sequence of events within each ADL. Likewise, intrinsic factors of users may influence the ADL duration. For example, some subjects may need more training than others before performing the simulations in a natural way. Additionally, the participants may prepare meals in multiple manners and ADL durations could therefore evidence high variability. Likewise, external aspects (i.e., the time of the day) may also contribute to this difference. In fact, some simulators have been reported as incapable of capturing the natural variations of activity durations in the presence of these elements [60]. Another factor potentially influencing these outcomes is the dissimilarities between home layouts of real and simulated environments. The distances travelled by users through the avatar are not equivalent to those covered at the apartment, which may increase the deviations between the duration metrics.

On a different note, detailed and high-quality data is required for effectively training AR models so that they can identify behavioural patterns in the daily routine of PwD. Given the limitations of real data collection widely depicted in the introduction (see Section 1), training datasets are expected to be formed by synthetic observations mimicking the intrinsic properties of real data. In this regard, it is not advised to use data evidencing significant deviations from the real world; therefore, these results reveal the need for approximating these observations to those from the real smart homes to increase the prediction accuracy and other performance indicators of AR models. The data analysts are then recommended to use the PLSR equations to transform the data and then include them in the training datasets. If this is not carried out, the AR model will evidence poor performance in identifying significant changes in the duration of each ADL.

These results are comparable with those presented in Lee et al. [15]. Specifically, the authors used a stochastic analysis to evaluate the similarity of the synthetic datasets produced by Persim (3Da, 3Du, and 1.5) in contrast with those gathered from the real environment. The analysis was focused on the occurrence probability and the number of sensor events. In summary, the highest average similarity was found to be 81% (Persim 1.5). Similarly, Dahmen and Cook [7] used significance comparative analysis, Euclidean distance, and Dynamic Time Warping (DWT) to examine the realism of data derived from the SynSyn simulator. However, both assessments were made in general and not targeted to each ADL as our paper does. In fact, this study reveals great equivalence in *Stay in bed*, *Get out of home*, and *Get cold drink*, but significant disparities in the remaining ADLs, which may be hidden in a general analysis. Conversely, Alharbi et al. [14] employed the GAN-test method to verify how well the simulated samples mimic the real data distributions. In this case, F1 scores revealed perfect equivalence in all datasets and classes considering *run*, *still*, *walk*, bike, and bus activities. Nonetheless, our case study takes into account a larger number of ADLs that are also more complex to analyse. ADLs like *cook dinner, make breakfast,* and get hot drink are performed in several ways, which is proven by a variable number of sensor events and a plurality of event sequences. Additionally, Kamara-Esteban et al. [16] appraised the coherence of synthetic datasets with real measurements in both singleuser and multiple-user scenarios considering the frequency and activity duration of five ADLs (Preparing breakfast, Washing dishes, Shower, Preparing dinner, and Do laundry). The authors deployed probability distributions and confidence intervals to evaluate the similarity. Nevertheless, a summary *p*-value statistic was not provided per each ADL so that disparities can be further categorized as significant or non-significant, thereby supporting

the use of models approximating the synthetic observations to a more realistic behaviour as this study does. Other important related efforts described in Alshammari et al. [17], Park et al. [63], Synnott et al. [64], Ariani et al. [65], and McGlinn et al. [66] did not present any comparative analysis between synthetic and real data from ADLs in smart homes.

#### 5.2. Transforming Synthetic Data to Predict Real Activity Duration

Simulated data are expected to emulate the real-world behaviour of PwD and then be useful to train AR models capable of detecting deviations or change points explaining the disease progress. Nonetheless, it is evident from Section 5.1 that some ADLs (Use restroom, Make breakfast, Stay in the office, Get hot drink, and Cook dinner) are significantly different in terms of activity duration when performed in both real and synthetic smart environments. In similar contexts, while Kamara-Esteban et al. [16], Alharbi et al. [14], and Lee et al. [15] did not use any method to transform the synthetic data before including them in the training datasets, Dahmen and Cook [7] proposed the use of 12 state hidden Markov models and probabilistic modelling techniques to transform the data; however, this study did not evidence how the models may vary depending on the ADL nature. Being aware of this, this Subsection presents PLSR models transforming the synthetic data to better approximate the real-world activity durations corresponding to these ADLs. The usability of each model was further assessed considering the assumptions explained in Section 3. Valid models were then described in terms of predictive power and fit to then demonstrate their capability for generating intricate real data that effectively complement the small-size datasets (built from the subjects staying at HINT). All the models were run on a PC with a 2.90 GHz Intel(R) Core(TM) i5-9400 CPU processor.

## 5.2.1. Use Restroom

When modelling real activity duration, it is essential to identify synthetic factors highly contributing to the response variable. Table 2 depicts the ANOVA test deployed to this aim. It can be noted that Synthetic Activity Duration  $(SAD_{U_R})$  (*p*-value = 0.021), Number of Events per Pressure Sensor (NEPS<sub>*P*</sub>) (*p*-value = 0.040), and the two-order interaction  $(SAD_{U_R} * NEPS_P)$  (*p*-value = 0.003) were concluded to be statistically significant on *Y* and can therefore be considered in the PLSR model. In fact, the logarithmic polynomial model (Equation (8)) was concluded to be appropriate for analysing the variability of activity duration when subjects use the restroom (*p*-value = 0.000;  $\alpha = 0.05$ ).

Table 2. ANOVA results for the Use restroom PLSR model.

Source	DF	Contribution	Adj SS	Adj MS	F-Value	<i>p</i> -Value
Regression	3	98.32%	123.55	41.18	100.79	0.000
SAD <sub>UR</sub>	1	3.61%	4.54	4.54	11.11	0.021
$NEPS_P$	1	2.46%	3.09	3.09	7.50	0.040
$SAD_U * NEPS_P$	1	9.47%	11.90	11.90	29.12	0.003
Error	5	1.62%	2.043	0.4087		
Total	8	100%	125.60			

On the other hand, the model presented in Equation (8) has also been evaluated from the goodness-of-fit and predictive ability perspectives (Table 3). In this respect, an important metric to study is the standard error of the estimate (S) which, in this instance, indicates that the average distance of the original observations from the model zone is about 0.639 s. This result is near 0, thereby demonstrating a close match for the prediction interval. Meanwhile, Table 3 postulates an  $AdjR^2 = 97.40\%$  revealing a high explanatory power of the model proposed for *Use restroom*. Additionally, satisfactory predictive performance was proved through the *PRESS* statistic (4.51), which favours the use of this model in providing valid predictions that can complement the training datasets employed by the AR algorithms. Random noise was also discarded, as there is no meaningful difference between the  $AdjR^2$  and  $R^2$  (*pred*) (0.99%). From the scalability point of view, it is good to highlight that the computational run time for this model ranged from 0.630 to 0.690 s.

Table 3. Predictive ability and fit of *Use restroom* PLSR model.

S	$R^2$	Adj R <sup>2</sup>	PRESS	$R^2$ (Pred)
0.639	98.37%	97.40%	4.51	96.41%

$$\ln Y = 0.1132 * SAD_{UR} + 0.298 * NEPS_P - 0.0077 * SAD_{UR} * NEPS_P$$
(8)

Finally, the *DW* statistic was calculated to verify the interdependence assumption of residuals. As DW > DU (2.549 > 1.777), neither negative nor positive auto-correlation is evidenced in the group of random errors. In the meantime, homogeneity of variances was confirmed, thereby validating the adequacy of the PLSR model shown in Equation (8).

## 5.2.2. Make Breakfast

Table 4 presents the results of the ANOVA test performed to detect synthetic variables that can support the prediction of time spent by the subjects when making breakfast. It is good to highlight that  $NEPA_{MB}$  (*p*-value = 0.004) was identified as a good predictor because the number of tasks may vary from one person to the other as there are different ways of cooking; even without considering the ordering of events, which is intrinsic in each subject. Likewise, the significance of  $SAD_{MB}$  (*p*-value = 0.004), also reported in Dahmen and Cook [7], explains how the specific characteristics of PwD can be captured through a simulator (e.g., IESim) to then be transformed for representing the real behaviour. Interestingly, when coupling these terms  $SAD_{MB} * NEPA_{MB}$  (p-value = 0.014), it was further noticed how each ADL (in this case, make breakfast) is differently performed by each individual, thereby underpinning the finding postulated by Fortino et al. [67] regarding the wide range of PwD profiles that may be encountered in the real world. Learning about these ADLs appropriately is of utmost importance for modelling activity durations more accurately. In this instance, the final outcome is a quadratic polynomial expression (Equation (9)) capable of predicting the time invested by the inhabitants of the HINT smart home when preparing breakfast (p-value = 0.000).

Source	DF	SS	Contributio	n Adj SS	Adj MS	F-Value	<i>p</i> -Value
Regression	3	777.21	99.07%	777.72	259.24	178.08	0.000
$SAD_{MB}$	1	739.72	94.23%	37.46	37.46	25.73	0.004
NEPA <sub>MB</sub>	1	18.33	2.33%	37.96	37.96	26.08	0.004
SAD <sub>MB</sub> * NEPA <sub>MB</sub>	1	19.66	2.50%	19.66	19.66	13.51	0.014
Error Total	5 8	7.27 785.00	0.93% 100%	7.27	1.45		

Table 4. ANOVA results for the Make breakfast PLSR model.

The performance indicators illustrated in Table 5 further detail the characteristics of the model (Equation (9)) with regards to the ability to forecast the real activity duration of inhabitants when preparing breakfast based on the simulator outputs. The findings point out a very small divergence between the  $AdjR^2$  and  $R^2$  (*pred*) (1.07%), therefore discarding the presence of overfitting effects hindering the applicability of the model to complement the datasets used for training the associated AR algorithms. Notably, an excellent fit is provided by the attained quadratic expression as  $AdjR^2$  is over 90%. No less important are the low values of *S* (1.206) and *PRESS* (20.04), which are accounted by the overlapping of fitted data with the prediction interval.

Table 5. Predictive ability and fit of Make breakfast PLSR model.

S	$R^2$	Adj R <sup>2</sup>	PRESS	$R^2$ (Pred)
1.206	99.07%	98.52%	20.04	97.45%

$$\sqrt{Y} = 0.2567 * \text{SAD}_{MB} + 0.930 * \text{NEPA}_{MB} - 0.0249 * \text{SAD}_{MB} * \text{NEPA}_{MB}$$
(9)

Every PLSR model needs to be appraised in search of potential abnormalities or random errors that may invalidate the predictions made by the model in presence of new synthetic entries. Consequently, the randomness of residuals was scanned via the *DW* indicator at a 5% significance level. As the DW > DU (2.064 > 1.777), sufficient evidence has been collected to reject potential inter-dependencies among the residuals. Complementary to this result, no proof of unequal variances was detected in the sample which, in addition to the foregoing, confirms the practicality of the model in real applications including the one related to the AR models for *Prepare breakfast*. In this case, the computational time invested in this model fluctuated between 0.480 and 0.550 s.

#### 5.2.3. Stay in the Office

Another challenging ADL to model based on synthetic data is *Stay in the office*. The time invested by individuals staying at the smart homes may differ from one to the other based on a plethora of extrinsic and intrinsic factors. Despite this, as seen in other ADLs, the number of events per activity (NEPA<sub>SIO</sub>; *p*-value = 0.000) and synthetic activity duration (SAD<sub>SIO</sub>; *p*-value = 0.006) were held to be good regressors of the time invested by the individuals when staying in the office (Table 6). In fact, as also evidenced in other ADLs, the interplay between these simulated variables was also concluded as significant on the response variable Y and can be included in the predictive model (Equation (10)).

Source	DF	SS	Contribution	n Adj SS	Adj MS	F-Value	<i>p</i> -Value
Regression	2 1	801.53 702.81	95.76% 83.97%	801.53 333.92	400.76 333.91	67.80 56.49	0.000
$SAD_{SIO} *$ NEPA <sub>SIO</sub>	1	98.72	11.79%	98.72	98.72	16.70	0.006
Error Total	6 8	35.47 837.00	4.24% 100%	35.47	35.47		

Table 6. ANOVA results for the Stay in the office PLSR model.

The proposed equation (Equation (10)) satisfies all the conditions for further use in real AR applications. First, both  $AdjR^2$  and  $R^2$  (pred) are over 90%, thus indicating that this quadratic polynomial expression explains a great proportion of the activity duration variation in *Stay in the office* and in the meantime is excellent for making predictions of the response variable. This allows us to create an approximation of real data without a time-consuming recruitment process and substantial investments. Training AR classifiers required datasets with enough size so that the algorithms can effectively learn about the human behaviour in a particular ADL [68]. Nonetheless, it is advised to collect a pre-sample of real data (approximately 10 individuals) to estimate the effects of new office layouts and other critical factors on the activity duration. In the near future, other synthetic features may be explored to increase the fitting and prediction power of the model. On a different tack, the small distance between  $AdjR^2$  and  $R^2$  (*pred*) (3.51%) proves that all of the terms do not substantially increase the model bias, which makes it useful for complementing the small-sized datasets constructed from the smart home environments. In addition to this body of findings, it is good to highlight the small values of S (2.431) and PRESS (64.83) that provide good support for the high prediction performance statement (Table 7). An important aspect is the run time, which varied from 0.580 to 0.610 s.

Table 7. Predictive ability and fit of *Stay in the office* PLSR model.

S	$R^2$	Adj R <sup>2</sup>	PRESS	$R^2$ (Pred)
2.431	95.76%	94.35%	64.83	92.25%

$$\sqrt{Y} = 2.227 * \text{NEPA}_{SIO} - 0.040 * \text{SAD}_{SIO} * \text{NEPA}_{SIO}$$
(10)

Ultimately, the presence of auto-correlation in the adjacent residuals of the PLSR model proposed for describing the duration of *Stay in the office* was evaluated via the Durbin–Watson test at a 5% significance level. This discards the fact that the PLSR equation can miscalculate the standard error of the coefficients and therefore show them significantly when they are really not. As D > DU (1.814 > 1.777), no correlation exists among the random errors. Furthermore, no heterogeneity was observed in the residuals, which favours the adequacy of the model when used for underpinning the training of the correspondent AR classifier.

# 5.2.4. Get Hot Drink

Table 8 presents the significance analysis of potential predictors included in Equation (11). In this instance, a *p*-value of 0.000 proves that the PLSR model proposed in Equation (11) is suitable for describing the variations of real activity duration *Y* when getting a hot drink (*p*-value = 0.000;  $\alpha$  = 0.05). This is supported features significantly contributing to the prediction of *Y*: Synthetic Activity Duration (SAD<sub>*GHD*</sub>) (*p*-value = 0.004), the Number of Events per Activity (NEPA<sub>*GHD*</sub>) (*p*-value = 0.004), and the interaction between these variables (SAD<sub>*GHD*</sub> \* NEPA<sub>*GHD*</sub>) (*p*-value = 0.002). All these synthetic factors were considered in a polynomial quadratic PLSR formula modelling the time spent by a person getting hot drinks at a real home environment.

Table 8. ANOVA results for the Get hot drink PLSR model.

Source	DF	SS	Contribution	Adj SS	Adj MS	F-Value	<i>p</i> -Value
Regression	3	1570.99	97.82%	1570.99	523.66	74.80	0.000
SĂD <sub>GHD</sub>	1	1287.84	80.19%	184.11	184.11	26.30	0.004
NEPA <sub>GHD</sub>	1	13.23	0.82%	174.71	174.71	24.95	0.004
SAD <sub>GHD</sub> * NEPA <sub>GHD</sub>	1	269.92	16.81%	269.92	269.92	38.55	0.002
Error	5	35.01	2.18%	35.01	7.001		
Total	8	1606.00	100%				

The model attained in Equation (11) evidences very good performance in both fit and predictive power domains (Table 9). First, the standard deviation *S* was estimated to be very small (2.645), and it can therefore be deemed as a symptom of model adequacy regarding the description of activity durations *Y* in *Get hot drink*. Moreover, Table 9 shows an  $AdjR^2 = 96.51\%$ , which demonstrates an excellent fit of Equation (11) with respect to the original observations. On a different tack, adequate predictive power was demonstrated as the *PRESS* indicator (77.27) was found to be very low. In a similar vein, an insignificant difference (1.32%) was noted between the  $AdjR^2$  and  $R^2$  (*pred*), thereby rejecting the overfitting problem in the model. From the applicability perspective, it is good to stress the low computational time invested by the data analysts to derive the model (0.580 to 0.690 s).

Table 9. Predictive ability and fit of Get hot drink PLSR model.

S	$R^2$	Adj R <sup>2</sup>	PRESS	$R^2$ (Pred)
2.645	97.82%	96.51%	77.27	95.19%

$$\sqrt{Y} = 0.2811 * \text{SAD}_{GHD} + 2.191 * \text{NEPA}_{GHD} - 0.0386 * \text{SAD}_{GHD} * \text{NEPA}_{GHD}$$
 (11)

The usability of the PLSR model proposed in Equation (11) is validated through the randomness and equal variance assumption (Equation (3)). The interdependence hypothesis was evaluated by employing the Durbin–Watson test (Equation (2)). In this case, the observed DW (2.981) was concluded to be higher than the DU limit (1.777), and it is thus not evidenced by the presence of significant auto-correlations among the random errors of the model. In the meantime, the homogeneity assumption was assessed; however, no sufficient support was detected against the null hypothesis. The above-mentioned reasons then provide a statistical base underpinning the model deployment in the real world.

#### 5.2.5. Cook Dinner

The ANOVA test (Table 10) indicates that the PLSR model proposed in Equation (12) is adequate to explain the variability of real activity duration *Y* when cooking dinner (*p*-value = 0.000;  $\alpha = 0.05$ ). More specifically, the Synthetic Activity Duration (SAD<sub>CD</sub>) (*p*-value = 0.002), the Number of Events per Chair Pressure Sensor (NEPS<sub>CHP</sub>) (*p*-value = 0.001), and the interaction between these factors (SAD<sub>CD</sub> \* NEPS<sub>CHP</sub>) (*p*-value = 0.007) were concluded to be good predictors of *Y* and can therefore be employed for modelling this response variable. In this instance, a logarithmic PLSR expression was found to be the most effective alternative for describing the time period taken by each subject when preparing dinner. It took from 0.510 to 0.690 s to generate the PLSR model.

Table 10. ANOVA results for the Cook dinner PLSR model.

Source	DF	SS	Contribution	n Adj SS	Adj MS	F-Value	<i>p</i> -Value
Regression	3	188.24	98.59%	188.24	62.74	116.28	0.000
SAD <sub>CD</sub>	1	163.16	85.45%	19.02	19.02	35.25	0.002
NEPS <sub>CHP</sub>	1	14.502	7.60%	24.28	24.28	45.00	0.001
SAD <sub>CD</sub> * NEPS <sub>CHP</sub>	1	10.57	5.54%	10.57	10.57	19.59	0.007
Error	5	2.698	1.41%	2.69	0.53		
Total	8	190.93	100%				

Upon assessing the predictive ability and fit inherent to the model in Equation (12), it is worth noting that the standard deviation of the data points around the fitted observations *S* is close to 0 (0.734 s), which demonstrates that there are no significant distances between the original *Y* values and the fitted data. We can accordingly postulate that the model (Equation (12)) is appropriate to describe the activity duration *Y* patterns corresponding to *Cook dinner*. Additionally, as depicted in Table 11, a high proportion of the variation in *Y* ( $AdjR^2 = 97.74\%$ ) is described by Equation (12), thereby providing good support for the goodness-of-fit hypothesis. On a different note, the *PRESS* statistic (13.55) was calculated to be small, thereby underpinning a satisfactory predictive ability of the model. Similarly, the predictive  $R^2$  (*pred*) (93.11%) proves that the polynomial model predicts the response for new observations at high performance and can be hence categorized with excellent predictive ability. No less important is the small gap between the  $AdjR^2$ and  $R^2$  (*pred*) (4.63%), which discards overfitting problems and confirms that the model is effective for estimating new real durations of *Cook dinner* activity when entering new SAD<sub>CD</sub> and NEPS<sub>CHP</sub> values into Equation (12).

Table 11. Predictive ability and fit of Cook dinner PLSR model.

s	$R^2$	Adj R <sup>2</sup>	PRESS	R <sup>2</sup> (Pred)
0.734	98.59%	97.74%	13.15	93.11%

 $\ln Y = 0.1888 * SAD_{CD} + 1.255 * NEPS_{CHP} - 0.0311 * SAD_{CD} * NEPS_{CHP}$ (12)

Lately, the independence assumption was verified through the Durbin–Watson (DW) statistic (Equation (2)). As the number of regressors denoted by k' = 2 (SAD<sub>CD</sub> and

NEPS<sub>*CHP*</sub>) and the sample size n = 8, the respective theoretical printed bounds (without intercept) are dL = 0.371 and dU = 1.777. Comparing the observed DW value (1.888) with the parameters dL and dU at the 5% level of significance, it is proven that no positive/negative auto-correlation exists among the model residuals (DW > dU). Likewise, no heterogeneity of residuals was detected in this case, thereby validating the suitability of the proposed model.

#### 6. Conclusions

Monitoring the daily living of PwD along the time supports the early diagnosis and treatment by healthcare systems, which contributes to increasing the life quality of these patients while alleviating the economic burden caused by this disease. The cornerstone is the effective automatic recognition of ADLs whose intrinsic properties and evolution provide a comprehensive overview of how dementia has progressed over time. Nonetheless, the performance of these recognition models is greatly based on the availability of intricate and suitable data. Adversely, there is accompanying difficulty in achieving formal ethical approval for gathering massive amounts of data while recruitment processes are costly, resource-consuming, and characterized by low engagement rates.

Simulation tools have arisen as a more efficient solution to obtaining synthetic data complementing the real datasets. The principal barrier hindering the potential applicability of these simulators is that synthetic data for critical behavioural variables (e.g., activity duration) have been concluded to be substantially divergent from the real observations in ADLs with a large number of tasks. This outcome is underpinned by differences in user profiles, home layouts, and background in the use of simulators. In this respect, significant dissimilarities were detected between the real and simulated activity durations of *Use restroom, Make breakfast, Stay in the office, Get hot drink,* and *Cook dinner*. In this case, the "activity duration" variable has been targeted considering its capability of evidencing behavioural changes in PwD, which may be an indicator of health decline as a result of a neurological deterioration process.

In this paper, we have proposed the use of PLSR models to transform the synthetic variables derived from the IE Sim simulator into a better approximation of real activity durations calculated from stays at the HINT smart home. It has aimed to bridge the previous gaps regarding research contrasting real and synthetic data. It goes beyond other reported approaches such as the SynSys algorithm [7], which does not consider the particular nature of each ADL, an aspect addressed by the proposed PLSR models.

Two primary contributions stem from this research. The main outcome has been the creation of predictive models providing a more accurate transformation of simulated data for describing real activity durations. In particular, logarithmic and quadratic polynomial models were defined for predicting the real duration of the aforementioned ADLs. All the models were found to provide good fitting ( $AdjR^2 > 90\%$ ) and predictive power  $(R^2 (pred) > 90\%; MaxS < 3; MaxPRESS < 100)$  whilst satisfying the PLSR adequacy assumptions. Therefore, they can be used for complementing the small-sized datasets employed for training the AR algorithms. A secondary contribution of this work has been the identification of the main synthetic predictors for real activity duration in each ADL, which helps to elucidate the intrinsic properties of each activity and how this metric may vary from one user to the other. Specifically, the Synthetic Activity Duration, Number of Events per Activity, Number of Events per Sensor per Activity, and their interactions were concluded to be the most popular regressors of real activity durations since they capture the intrinsic properties of human behaviour when staying at home. Of course, the results presented here may be limited to the specific layouts of IESim and HINT, as well as the experience of the inhabitants in the use of the simulator. Hence, it is recommended to first train the users on how to perform the ADLs virtually until overcoming the learning curve.

Future works will consider the sequence of tasks (sensor activations) performed by the subjects within each ADL. Thereby, it will be possible to discriminate and model the intrinsic properties of human behaviour. In a similar vein, it is important to explore how the sensor event ordering can be included in the PLSR models and whether the predictive ability and fit can be upgraded for producing a better approximation of the real activity duration. Another pathway to investigative is the application of other prediction algorithms, including Random Forest and Naïve Bayes, to perform comparative analysis with a focus on the ADLs with significant differences between real and synthetic activity durations as identified in this work.

Author Contributions: Conceptualization, M.O.-B. and E.J.; methodology, M.O.-B. and E.J.; software, M.O.-B., N.J.-R. and S.A.-F.; validation, M.O.-B. and E.J.; formal analysis, M.O.-B., N.J.-R., S.A.-F. and E.J.; investigation, M.O.-B., E.J., M.G.-C., C.N., I.C. and S.A.-F.; resources, C.N.; data curation, M.O.-B., N.J.-R., S.A.-F. and E.J.; writing—original draft preparation, M.O.-B., E.J., M.G.-C., I.C., N.J.-R. and S.A.-F.; writing—review and editing, M.O.-B., E.J., M.G.-C., C.N., I.C. and S.A.-F.; visualization, M.O.-B., N.J.-R., S.A.-F. and E.J.; supervision, M.O.-B. and C.N.; project administration, C.N.; funding acquisition, C.N. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research has received funding under the REMIND project Marie Sklodowska-Curie EU Framework for Research and Innovation Horizon 2020, under Grant Agreement No. 734355.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors thank to Jonathan Synnott for his previous work in the field.

Conflicts of Interest: The authors declare no conflict of interest.

#### Abbreviations

The following abbreviations are used in this manuscript:

ADLs	Activities of Daily Living
AI	Artificial Intelligence
ANOVA	Analysis of Variance
AR	Activity Recognition
CNN	Convolutional Neural Network
DF	Degrees of Freedom
DTW	Dynamic Time Warping
DW	Durbin Watson
GANs	Generative Adversarial Networks
HINT	Halmstad Intelligent Home
HMM	Hidden Markov Model
ML	Machine Learning
MS	Mean Square
NEA	Number of Events per Activity
NEPA <sub>GHD</sub>	Number of Events per Activity for Get Hot Drink
NEPA <sub>MB</sub>	Number of Events per Activity for Make Breakfast
NEPA <sub>SIO</sub>	Number of Events per Activity for Stay in the Office
$NEPS_P$	Number of Events per Pressure Sensor
NEPS <sub>CHP</sub>	Number of Events per Sensor (Chair Pressure)
NESA	Number of Events per Sensor per Activity
NIPALS	Nonlinear Iterative Partial Least Squares
OLSR	Ordinary Least Squares Regression (OLSR)
PCA	Principal Components Analysis
PCR	Principal Components Regression
PIR	Passive Infrared
PLS	Partial Least Squares
PLSR	Partial Least Squares Regression

PRESS	Prediction Residuals Sum of Squares
PwD	People with Dementia
SAD	Smoking Activity Dataset
SAD <sub>CD</sub>	Synthetic Activity Duration for Cook Dinner
SAD <sub>GHD</sub>	Synthetic Activity Duration for Get Hot Drink
$SAD_{MB}$	Synthetic Activity Duration for Make Breakfast
SAD <sub>SIO</sub>	Synthetic Activity Duration for Stay in the Office
SAD <sub>UR</sub>	Synthetic Activity Duration for Use Restroom
SHL	Sussex-Huawei Locomotion
SS	Squared Sum
UTI	Urinary Tract Infection
WHO	World Health Organization

#### References

- 1. World Health Organisation Dementia Key Facts. Available online: https://www.who.int/news-room/fact-sheets/detail/ dementia (accessed on 28 April 2022).
- 2. McConaghy, R.; Caltabiano M.L. Caring for a person with dementia: Exploring relationships between perceived burden, depression, coping and well-being. *Nurs. Health Sci.* 2005, *7*, 81–91. [CrossRef] [PubMed]
- 3. Lorenz, K.; Freddolino, P.P.; Comas-Herrera, A.; Knapp, M.; Damant, J. Technology-based tools and services for people with dementia and carers: Mapping technology onto the dementia care pathway. *Dementia* **2019**, *18*, 725–741. [CrossRef] [PubMed]
- 4. Garcia-Constantino, M.; Orr, C.; Synnott, J.; Shewell, C.; Ennis, A.; Cleland, I.; Nugent, C.; Rafferty, J.; Morrison, G.; Larkham, L.; et al. Design and Implementation of a Smart Home in a Box to Monitor the Wellbeing of Residents with Dementia in Care Homes. *Front. Digit. Health* **2021**, *3*, 798889. [CrossRef] [PubMed]
- Ortíz-Barrios, M.A.; Garcia-Constantino, M.; Nugent, C.; Alfaro-Sarmiento, I. A Novel Integration of IF-DEMATEL and TOPSIS for the Classifier Selection Problem in Assistive Technology Adoption for People with Dementia. *Int. J. Environ. Res. Public Health* 2022, 19, 1133. [CrossRef]
- Ortíz-Barrios, M.A.; Lundström, J.; Synnott, J.; Järpe, E.; Sant'Anna, A. Complementing real datasets with simulated data: A regression-based approach. *Multimed. Tools Appl.* 2020, 79, 34301–34324. [CrossRef]
- 7. Dahmen, J.; Cook, D. SynSys: A synthetic data generation system for healthcare applications. Sensors 2019, 19, 1181. [CrossRef]
- 8. HekmatiAthar, S.; Goins, H.; Raymond, S.; Byfield, G.; Anwar, M. Data-driven forecasting of agitation for persons with dementia: A deep learning-based approach. *SN Comput. Sci.* **2021**, *2*, 326. [CrossRef]
- Urwyler, P. Stucki, R.; Rampa, L.; Müri, R.; Mosimann, U.P.; Nef, T. Cognitive impairment categorized in community-dwelling older adults with and without dementia using in-home sensors that recognise activities of daily living. *Sci. Rep.* 2017, 7, 42084. [CrossRef]
- 10. Damla, A.; Wang, Y.; Bouchachia, A. Detection of dementia-related abnormal behaviour using recursive auto-encoders. *Sensors* **2021**, *21*, 260. [CrossRef]
- Enshaeifar, S.; Zoha, A.; Skillman, S.; Markides, A.; Acton, S.T.; Elsaleh, T.; Kenny, M.; Rostill, H.; Nilforooshan, R.; Barnaghi, P. Machine learning methods for detecting urinary tract infection and analysing daily living activities in people with dementia. *PLoS ONE* 2019 14, e0209909. [CrossRef]
- Virone, G.; Lefebvre, B.; Noury, N.; Demongeot, J. Modeling and computer simulation of physiological rhythms and behaviors at home for data fusion programs in a telecare system. In Proceedings of the 5th International Workshop on Enterprise Networking and Computing in Healthcare Industry, Santa Monica, CA, USA, 7 June 2003; pp. 111–117.
- Helal, A.; Mendez-Vazquez, A.; Hossain, S. Specification and synthesis of sensory datasets in pervasive spaces. In Proceedings of the 2009 IEEE Symposium on Computers and Communications, Sousse, Tunisia, 5–8 July 2009; pp. 920–925.
- Alharbi, F.; Ouarbya, L.; Ward, J.A. Synthetic sensor data for human activity recognition. In Proceedings of the 2020 International Joint Conference on Neural Networks (IJCNN), Glasgow, UK, 19–24 July 2020; IEEE: Piscataway, NJ, USA, 2020; pp. 1–9.
- 15. Lee, J.W.; Cho, S.; Liu, S.; Cho, K.; Helal, S. Persim 3d: Context-driven simulation and modeling of human activities in smart spaces. *IEEE Trans. Autom. Sci. Eng.* **2015**, *12*, 1243–1256. [CrossRef]
- 16. Kamara-Esteban, O.; Azkune, G.; Pijoan, A.; Borges, C.E.; Alonso-Vicario, A.; López-de-Ipiña, D. MASSHA: An agent-based approach for human activity simulation in intelligent environments. *Pervasive Mob. Comput.* **2017**, *40*, 279–300. [CrossRef]
- 17. Alshammari, N.; Alshammari, T.; Sedky, M.; Champion, J.; Bauer, C. Openshs: Open smart home simulator. *Sensors* **2017**, *17*, 1003. [CrossRef] [PubMed]
- 18. Damien, B.; Nguyen, S.M.; Lohr, C.; LeDuc, B.; Kanellos, I. A survey of human activity recognition in smart homes based on IoT sensors algorithms: Taxonomies, challenges, and opportunities with deep learning. *Sensors* **2021**, *21*, 6037. [CrossRef]
- 19. Wang, K.F.; Gou, C.; Zheng, N.N.; Rehg, J.M.; Wang, F.-Y. Parallel vision for perception and understanding of complex scenes: Methods, framework and perspectives. *Artif. Intell. Rev.* 2017, *48*, 299–329. [CrossRef]
- Reeves, D.R.; Taylor, S.J. Selection of training data for neural networks by a genetic algorithm. In Proceedings of the International Conference on Parallel Problem Solving from Nature 1998, Amsterdam, The Netherlands, 27–30 September 1998; Springer: Berlin/Heidelberg, Germany, 2006. [CrossRef]

- 21. Kleijnen, J.P.C. Verification and validation of simulation models. Eur. J. Oper. Res. 1995, 82, 145–162. [CrossRef]
- 22. Wold, H. Estimation of principal components and related models by iterative least squares. In *Multivariate Analysis*; Academic Press: Cambridge, MA, USA, 1966; pp. 391–420.
- 23. Otto, M.; Wegscheider, W. Selectivity in multicomponent analysis. Anal. Chim. Acta 1986, 180, 445–456. [CrossRef]
- 24. Pirouz, D.M. An Overview of Partial Least Squares. In *ERN: Other Econometrics: Econometric & Statistical Methods (Topic)*; SSRN: Rochester, NY, USA, 2006. [CrossRef]
- Wold, S.; Ruhe, A.; Wold, H.; Dunn, W.J., III. The Collinearity Problem in Linear Regression. The Partial Least Squares (PLS) Approach to Generalized Inverses. SIAM J. Sci. Stat. Comput. 1984, 5, 735–743. [CrossRef]
- Wold, S.; Albano, C.; Dunn, W.J., III; Edlund, U.; Esbensen, K.; Geladi, P.; Hellberg, S.; Johansson, E.; Lindeberg, W.; Sjöström, M. Multivarite Data Analysis in Chemistry. In *Chemometrics: Mathematics and Statistics*; Kowalski, B., Ed.; D. Riedel Publishing Company: Dordrecht, The Netherlands, 1984; Volume 17.
- Wold, S.; Albano, C.; Dunn, W.J., III; Esbensen, K.; Hellberg, S.; Johansson, E.; Sjöström, M.; Martens, H.; Rosswurm, J. Food Research and Data Analysis; Martens, W., Rosswurm, H., Eds.; Applied Science Publishers: London, UK, 1984.
- 28. Geladi, P.; Kowalski, B. Partial least-squares regression—A tutorial. Anal. Chim. Acta 1986, 185, 1–17. [CrossRef]
- 29. Camarrone, F.; van Hulle, M.M. Fast Multiway Partial Least Squares Regression. IEEE Trans. Biomed. Eng. 2019, 66, 433. [CrossRef]
- 30. Henseler, J. On the convergence of the partial least squares path modeling algorithm. Comput. Stat. 2010, 25, 107–120. [CrossRef]
- Jiang, X.B.; You, X.G.; You, S.J.; Tao, D.C.; Chen, C.L.P.; Cheung, Y.M. Variance constrained partial least squares. *Chemom. Intell. Lab. Syst.* 2015, 145, 60–71. [CrossRef]
- Mou, Y.; Zhou, L.; You, X.G.; Lu, Y.L.; Chen, W.Z.; Zhao, X. Multiview Partial Least Squares. Chemom. Intell. Lab. Syst. 2017, 160, 13–21. [CrossRef]
- Singer, M.; Krivobokova, T.; Munk, A.; Groot, B. Partial least squares for dependent data. *Biometrika* 2016, 103, 351–362. [CrossRef]
   [PubMed]
- 34. Helland, I.S.; Saebo, S.; Almoy, T.; Rimal, R. Model and estimators for partial least squares regression. J. Chemom. 2018, 32, e3044.
- 35. Butler, N.A.; Denham, M.C. The peculiar shrinkage properties of partial least squares regression. *J. R. Stat. Soc. B* 2000, *62*, 585–593. [CrossRef]
- Hinkle, J.; Raynes, W. Partial least squares and compositional data: Problems and alternatives. *Chemom. Intell. Lab. Syst.* 1995, 30, 159–172. [CrossRef]
- Rönkkö, M.; McIntosh, C.N.; Antonakis, J.; Edwards, J.R. Partial least squares path modeling: Time for some serious second thoughts. J. Oper. Manag. 2016, 47–48, 9–27. [CrossRef]
- 38. Chui, K.T.; Lytras, M.D.; Vasant, P. Combined Generative Adversarial Network and Fuzzy C-Means Clustering for Multi-Class Voice Disorder Detection with an Imbalanced Dataset. *Appl. Sci.* 2020, *10*, 4571. [CrossRef]
- Hong, Y.; Hwang, U.; Yoo, J.; Yoon, S. How Generative Adversarial Networks and Their Variants Work: An Overview. ACM Comput. Surv. 2019, 52, 1–43. [CrossRef]
- 40. Alqahtani, H.; Kavakli, M.; Kumar, G. Applications of Generative Adversarial Networks (GANs): An Updated Review. *Arch. Comput. Methods Eng.* **2021**, *28*, 525–552. [CrossRef]
- Hui, J. GAN—Why It Is So Hard to Train Generative Adversarial Networks! Medium. 2018. Available online: https://jonathanhui.medium.com/gan-why-it-is-so-hard-to-train-generative-advisory-networks-819a86b3750b (accessed on 1 July 2022).
- 42. Luo, F.-L. Machine Learning for Future Wireless Communications; John Wiley & Sons: Hoboken, NJ, USA, 2020. [CrossRef]
- Zeng, X.-Q.; Li, G.-Z. Incremental partial least squares analysis of big streaming data. *Pattern Recognit.* 2014, 47, 3726–3735. [CrossRef]
- 44. Kearns, M.J. The Computational Complexity of Machine Learning; MIT Press: Cambridge, MA, USA, 1990. 76476. [CrossRef]
- Tang, Z.; Luo, L.; Xie, B.; Zhu, Y.; Zhao, R.; Bi, L.; Lu, C. Automatic sparse connectivity learning for neural networks. *IEEE Trans. Neural Netw. Learn. Syst.* 2022, 1–15. [CrossRef] [PubMed]
- Huang, Q. Weight-Quantized SqueezeNet for Resource-Constrained Robot Vacuums for Indoor Obstacle Classification. *Artif.* Intell. 2022, 3, 180–193. [CrossRef]
- Chen, C.; Cao, X.; Tian, L. Partial least squares regression performs well in MRI-based individualized estimations. *Front. Neurosci.* 2019, 13, 1282. [CrossRef]
- Hansen, P.M.; Schjoerring, J.K. Reflectance measurement of canopy biomass and nitrogen status in wheat crops using normalized difference vegetation indices and partial least squares regression. *Remote Sens. Environ.* 2003, 86, 542–553. [CrossRef]
- Hao, C.; Chen, D. Software/Hardware Co-design for Multi-modal Multi-task Learning in Autonomous Systems. In Proceedings of the IEEE 3rd International Conference on Artificial Intelligence Circuits and Systems (AICAS), Washington, DC, USA, 6–9 June 2021. [CrossRef]
- 50. Zheng, J.; Lu, C.; Hao, C.; Chen, D.; Guo, D. Improving the Generalization Ability of Deep Neural Networks for Cross-Domain Visual Recognition. *IEEE Trans. Cogn. Dev. Syst.* 2021, 13, 607–620. [CrossRef]
- 51. Allen, D.M. The Relationship Between Variable Selection and Data Agumentation and a Method for Prediction. *Technometrics* **1974**, *16*, 125–127. [CrossRef]
- Martínez, J.L.; Saulo, H.; Barrios Escobar, H.; Leão, J. A new model selection criterion for partial least squares regression. *Chemom. Intell. Lab. Syst.* 2017, 169, 64–78. [CrossRef]

- 53. Liu, H.; Weiss, R.E.; Jennrich, R.I.; Wenger, N.S. PRESS model selection in repeated measures data. *Comput. Stat. Data Anal.* **1999**, 30, 169–184. [CrossRef]
- 54. Pratt, J.W.; Gibbons, J.D. Concepts of Nonparametric Theory; Springer: New York, NY, USA, 1981. [CrossRef]
- 55. Kolassa, J.D. Edgeworth approximations for rank sum test statistics. Stat. Probab. Lett. 1995, 24, 169–171. [CrossRef]
- 56. Cox, D.R.; Hinkley, D.V. Theoretical Statistics; Chapman & Hall: New York, NY, USA, 1974. [CrossRef]
- 57. Larsen, R.J.; Marx, M.L. An Introduction to Mathematical Statistics and Its Applications, 6th ed.; Pearson: Boston, MA, USA, 2018. [CrossRef]
- Lundström, J.; Morais, W.O.D.; Menezes, M.; Gabrielli, C.; Bentes, J.; Sant'Anna, A.; Synnott, J.; Nugent, C. Halmstad intelligent home-capabilities and opportunities. In Proceedings of the International Conference on IoT Technologies for HealthCare, Budapest, Hungary, 18–19 October 2016; Springer: Cham, Switzerland, 2016; pp. 9–15. [CrossRef]
- Hamad, R.A.; Järpe, E.; Lundström, J. Stability analysis of the t-SNE algorithm for human activity pattern data. In Proceedings of the 2018 IEEE International Conference on Systems, Man, and Cybernetics (SMC), Miyazaki, Japan, 7–10 October 2018; IEEE: Piscataway, NJ, USA, 2018; pp. 1839–1845.
- 60. Synnott, J.; Nugent, C.; Jeffers, P. Simulation of smart home activity datasets. Sensors 2015, 15, 14162–14179. [CrossRef] [PubMed]
- 61. Chen, K.; Zhang, D.; Yao, L.; Guo, B.; Yu, Z.; Liu, Y. Deep learning for sensor-based human activity recognition: Overview, challenges, and opportunities. *ACM Comput. Surv. CSUR* **2021**, *54*, 1–40. [CrossRef]
- 62. Katz, S. Assessing self-maintenance: Activities of daily living, mobility, and instrumental activities of daily living. *J. Am. Geriatr. Soc.* **1983**, *31*, 721–727. [CrossRef] [PubMed]
- Park, B.; Min, H.; Bang, G.; Ko, I. The User Activity Reasoning Model in a Virtual Living Space Simulator. *Int. J. Softw. Eng. Appl.* 2015, 9, 53–62. [CrossRef]
- Synnott, J.; Chen, L.; Nugent, C.D.; Moore, G. The creation of simulated activity datasets using a graphical intelligent environment simulation tool. In Proceedings of the 2014 36th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Chicago, IL, USA, 26–30 August 2014; IEEE: Piscataway, NJ, USA, 2014; pp. 4143–4146.
- Ariani, A.; Redmond, S.J.; Chang, D.; Lovell, N.H. Simulation of a smart home environment. In Proceedings of the 2013 3rd International Conference on Instrumentation, Communications, Information Technology and Biomedical Engineering (ICICI-BME), Bandung, Indonesia, 7–8 November 2013; IEEE: Piscataway, NJ, USA, 2013; pp. 27–32.
- 66. McGlinn, K.; O'Neill, E.; Gibney, A.; O'Sullivan, D.; Lewis, D. SimCon: A Tool to Support Rapid Evaluation of Smart Building Application Design using Context Simulation and Virtual Reality. *J. Univers. Comput. Sci.* **2010**, *16*, 1992–2018.
- 67. Fortino, G.; Guzzo, A.; Ianni, M.; Leotta, F.; Mecella, M. Predicting activities of daily living via temporal point processes: Approaches and experimental results. *Comput. Electr. Eng.* **2021**, *96*, 107567. [CrossRef]
- 68. Saleh, M.; Abbas, M.; Le Jeannes, R.B. FallAllD: An open dataset of human falls and activities of daily living for classical and deep learning applications. *IEEE Sens. J.* 2020, *21*, 1849–1858. [CrossRef]